Author's response to reviews

Title: Prioritising neonatal medicines research: UK Medicines for Children Research Network scoping survey

Authors:

Mark A Turner (mark.turner@liverpool.ac.uk)
Sara Lewis (sara.lewis@npeu.ox.ac.uk)
Dan B Hawcutt (dhawcutt@liv.ac.uk)
D Field (david.field@uhl-tr.nhs.uk)

Version: 7 Date: 11 August 2009

Author's response to reviews:

I have made minor corrections to ensure the numbers in the text are correct and match the numbers in the Tables.

Thank you,
m.